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ABSTRACT 

Energy is the most prominent need of today's world, and developing sustainable and reliable 

energy sources is one of the major challenges.  Due to an increase in the global population, the 

demand for clean energy sources, i.e., sources with a minimum negative impact on the 

environment is continuously increasing. Contemporary research has focused on developing 

energy sources that reduce fossil fuel consumption without creating adverse environmental 

effects. Hydrogen energy seems to be the most appropriate solution due to its high energy 

density and clean-burning ability. The production of green hydrogen (i.e., from renewable 

electricity) and blue hydrogen (i.e., from fossil fuel with carbon capture) are most explored 

nowadays due to environmental concerns.  

Due to a lack of infrastructure and technological advancement, green hydrogen production is 

not currently economically feasible. This work aims to develop a technology that can produce 

blue hydrogen and has the potential to compete with commercial hydrogen production 

technologies. The methane decomposition (MD) process has the potential to produce blue 

hydrogen by capturing carbon as valuable carbon nanotubes. As one of the products is valuable 

carbon nanotubes (CNTs), the overall feasibility of the process can be governed by the quality 

and the cost of produced CNT. A series of catalysts were prepared from transition metals such 

as Ni and Fe using Cu, Zn as promoters, and alumina as support. A fluidized bed reactor was 

utilized for the methane decomposition reaction considering commercialization aspects. The 

catalysts were prepared by the wet-impregnation method, and their hydrodynamic behavior 

was studied by cold flow studies. The catalysts were characterized by various analytical 

techniques such as X-ray diffraction (XRD), temperature-programmed reduction (TPR), 

surface area analysis (BET-BJH), field-emission scanning electron microscopy (FESEM), 
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high-resolution transmission electron microscopy (HRTEM), atomic force microscopy (AFM), 

Raman spectroscopy, and thermogravimetric analysis (TGA) to comprehend the properties of 

catalysts (calcined, spent) and produced carbon materials (i.e., carbon nanotubes, graphene 

grafted CNTs). 

The performance of the catalysts was evaluated in the bubbling fluidized bed reactor over a 

wide range of operating conditions. A hydrodynamic study was carried out to identify the 

catalyst particle size and other parameters such as minimum fluidization velocity (Umf) and 

superficial velocity. A 60%Ni-5%Cu-5%Zn/Al2O3 catalyst with 125 µm (avg.) particle size 

provided more than 90% initial methane conversion at 750 °C. The produced carbon was in 

the form of valuable carbon nanotubes with a length up to 5 µm, outside diameter of 60-80 nm, 

and Raman ratio (i.e., ID/IG) of 0.5. The produced CNTs were extracted from the spent catalyst 

by a newly developed non-destructive ultrasonication process. The reusability of the catalyst 

was checked by performing five reaction-regeneration cycles, including carbon nanotube 

extraction by ultrasonication. It was observed that 60%Ni-5%Cu-5%Zn/Al2O3 catalyst 

regained its initial activity after each regeneration cycle without any significant loss in activity. 

A total of five reaction-regeneration cycles were performed, and the catalyst was able to 

provide >90% initial methane conversion after the fifth regeneration cycle. After each reaction 

cycle, the produced carbon nanotubes were characterized by various characterization tools, and 

no degradation in the CNT quality was observed.  

In order to observe the continuous change in hydrodynamic behavior of the catalyst due to 

CNT deposition, a hydrodynamic study was performed in the fluidized bed reactor at reaction 

conditions on different amounts of CNT deposited catalysts. Based on the latest literature, an 

economic analysis was carried out to find the break-even point between SMR and MD process. 
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The selling cost of the carbon nanotube was predicted by comparing the physical properties 

between produced and commercial carbon nanotubes. Efforts were made to reduce the cost of 

the catalyst by lowering the active metal content. A 20%Ni-1.7%Cu-1.7%Zn/Al2O3 catalyst 

provided >80% initial methane conversion, which remained stable up to 2h (>70%) with a high 

amount of CNTs (i.e., 18 gCNT gmetal
-1). The effect of metal-support interaction on hydrogen 

yield and carbon nanotube growth mechanism was also studied. 

The produced carbon nanotubes were also utilized as an autocatalyst for methane 

decomposition reaction to avoid the catalyst's regeneration. For that, 60%Ni-5%Cu-

5%Zn/Al2O3 and 30%Fe-2.5%Cu-2.5%Zn/Al2O3 catalysts were tested in a 750-950 °C 

temperature range. At 900 °C, after deposition of 50 g CNTs, 60%Ni-5%Cu-5%Zn/Al2O3 

catalyst provided 30% constant conversion for 40 h without any deactivation. The product gas 

with 30% H2-70% CH4 (called hythane gas) can be used as a potential fuel for an automotive 

application. While using produced carbon nanotube as an autocatalyst, a new form of carbon 

(i.e., graphene grafted carbon nanotube) was identified. Its growth mechanism and physical 

properties were explored by various analytical techniques such as FESEM, EDX mapping, 

HRTEM, Raman spectroscopy, and TGA. It was observed that graphene grafted CNTs are 

highly crystalline and more thermal stable than produced carbon nanotubes. Further, the 

methane decomposition mechanism on existing CNTs was studied and explained by various 

analytical tools. The produced carbon nanotube was utilized to produce ultra-light, flexible, 

highly porous aerogel materials. Prepared silica-CNT-CMC aerogel was utilized in an oil 

adsorption application, and 1 g carbon nanotube aerogel can adsorb 28 g of oil. 
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सार 
 ऊर्ाा आर् की दनुिया की सबसे प्रमुख र्रूरत है, और स्थायी और विश्िसिीय ऊर्ाा 
स्रोतों का विकास प्रमुख चुिौनतयों में से एक है। िैश्श्िक र्िसंख्या में िदृ्धि के कारण, स्िच्छ 
ऊर्ाा स्रोतों, यािी पयाािरण पर न्यूितम िकारात्मक प्रभाि िाले स्रोतों की मांग लगातार बढ़ 
रही है। समकालीि अिुसिंाि िे ऊर्ाा स्रोतों को विकससत करिे पर ध्याि कें द्रित ककया है र्ो 
प्रनतकूल पयाािरणीय प्रभाि पैदा ककए बबिा र्ीिाश्म ईंिि की खपत को कम करते हैं। हाइड्रोर्ि 
ऊर्ाा अपिे उच्च ऊर्ाा घित्ि और स्िच्छ र्लिे की क्षमता के कारण सबसे उपयुक्त समािाि 
प्रतीत होती है। पयाािरण संबंिी धचतंाओं के कारण आर्कल हररत हाइड्रोर्ि (अथाात, ििीकरणीय 
बबर्ली से) और िीले हाइड्रोर्ि (यािी, र्ीिाश्म ईंिि से काबाि कैप्चर के साथ) का उत्पादि 
सबसे अधिक ककया र्ाता है। 

 बुनियादी ढांचे और तकिीकी प्रगनत की कमी के कारण, ितामाि में हररत हाइड्रोर्ि 
का उत्पादि आधथाक रूप से संभि िहीं है। इस काया का उद्देश्य एक ऐसी तकिीक विकससत 
करिा है र्ो िीले हाइड्रोर्ि का उत्पादि कर सके और इसमें िाणणश्ययक हाइड्रोर्ि उत्पादि 
प्रौद्योधगककयों के साथ प्रनतस्पिाा करिे की क्षमता हो। मीथेि अपघटि (एमडी) प्रकिया में 
काबाि को मूल्यिाि काबाि िैिोट्यूब के रूप में कैप्चर करके िीले हाइड्रोर्ि का उत्पादि करिे 
की क्षमता है। चंूकक उत्पादों में से एक मूल्यिाि काबाि िैिोट्यूब (सीएिटी) है, प्रकिया की 
समग्र व्यिहायाता गुणित्ता और उत्पाद्रदत सीएिटी की लागत से नियंबित हो सकती है। उत्प्रेरकों 
की एक श्ृंखला Ni और Fe रै्से संिमण िातुओं से Cu, Zn को प्रिताक के रूप में, और एल्यूसमिा 
को समथाि के रूप में उपयोग करके तैयार की गई थी। व्यािसायीकरण पहलुओं पर विचार 
करते हुए मीथेि अपघटि प्रनतकिया के सलए एक ििीकृत बबस्तर ररएक्टर का उपयोग ककया 
गया था। उत्प्रेरक गीले-संसेचि विधि द्िारा तैयार ककए गए थे, और उिके हाइड्रोडायिासमक 
व्यिहार का अध्ययि शीत प्रिाह अध्ययि द्िारा ककया गया था। उत्प्रेरकों को एक्स-रे वििताि 
(एक्सआरडी), तापमाि-िमादेसशत कमी (टीपीआर), सतह क्षेि विश्लेषण (बीईटी-बीरे्एच), क्षेि-
उत्सर्ाि स्कैनिगं इलेक्रॉि माइिोस्कोपी (एफईएसईएम), उच्च-ररजॉल्यूशि रांससमशि इलेक्रॉि 
र्ैसी विसभन्ि विश्लेषणात्मक तकिीकों की विशेषता थी। माइिोस्कोपी (एचआरटीईएम), परमाणु 
बल माइिोस्कोपी (एएफएम), रमि स्पेक्रोस्कोपी, और थमोग्रैविमेद्ररक विश्लेषण (टीर्ीए) 
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उत्प्रेरक (कैलक्लाइंड, खचा ककए गए) और उत्पाद्रदत काबाि सामग्री (यािी, काबाि िैिोट्यूब, 

ग्राफीि ग्राफ्टेड सीएिटी) के गुणों को समझिे के सलए। 

 उत्प्रेरकों के प्रदशाि का मूल्यांकि बुदबुदाती ििीकृत बबस्तर ररएक्टर में पररचालि 
श्स्थनतयों की एक विस्ततृ श्ृंखला में ककया गया था। उत्प्रेरक कण आकार और अन्य पैरामीटर 
रै्से न्यूितम ििीकरण िेग (यूएमएफ) और सतही िेग की पहचाि करिे के सलए एक 
हाइड्रोडायिासमक अध्ययि ककया गया था। 125 माइिोि (औसत) कण आकार के साथ 
60%Ni-5%Cu-5%Zn/Al2O3 उत्प्रेरक िे 750 डडग्री सेश्ल्सयस पर 90% से अधिक प्रारंसभक 
मीथेि रूपांतरण प्रदाि ककया। उत्पाद्रदत काबाि मूल्यिाि काबाि िैिोट्यूब के रूप में था श्र्सकी 
लंबाई 5 माइिोि तक, बाहरी व्यास 60-80 एिएम और रमि अिुपात (यािी, आईडी / आईर्ी) 
0.5 था। उत्पाद्रदत सीएिटी को एक िि विकससत गैर-वििाशकारी अल्रासोनिकेशि प्रकिया 
द्िारा खचा ककए गए उत्प्रेरक से निकाला गया था। अल्रासोनिकेशि द्िारा काबाि िैिोट्यूब 
निष्कषाण सद्रहत पांच प्रनतकिया-पुिर्ािि चिों का प्रदशाि करके उत्प्रेरक की पुि: प्रयोययता 
की र्ााँच की गई थी। यह देखा गया कक 60%Ni-5%Cu-5%Zn/Al2O3 उत्प्रेरक िे गनतविधि 
में ककसी भी महत्िपूणा िुकसाि के बबिा प्रत्येक पुिर्ािि चि के बाद अपिी प्रारंसभक 
गनतविधि को पुिः प्राप्त कर सलया। कुल पांच प्रनतकिया-पुिर्ािि चि ककए गए, और उत्प्रेरक 
पांचिें पुिर्ािि चि के बाद >90% प्रारंसभक मीथेि रूपांतरण प्रदाि करिे में सक्षम था। 
प्रत्येक प्रनतकिया चि के बाद, उत्पाद्रदत काबाि िैिोट्यूब को विसभन्ि लक्षण िणाि उपकरणों 
द्िारा धचबित ककया गया था, और सीएिटी गुणित्ता में कोई धगरािट िहीं देखी गई थी। 

 सीएिटी र्मा होिे के कारण उत्प्रेरक के हाइड्रोडायिासमक व्यिहार में निरंतर पररिताि 
का निरीक्षण करिे के सलए, सीएिटी र्मा उत्प्रेरक की विसभन्ि मािाओं पर प्रनतकिया की 
श्स्थनत में िवित बबस्तर ररएक्टर में एक हाइड्रोडायिासमक अध्ययि ककया गया था। ििीितम 
साद्रहत्य के आिार पर, एसएमआर और एमडी प्रकिया के बीच ब्रेक-ईिि बबदं ुखोर्िे के सलए 
एक आधथाक विश्लेषण ककया गया था। उत्पाद्रदत और िाणणश्ययक काबाि िैिोट्यूब के बीच 
भौनतक गुणों की तुलिा करके काबाि िैिोट्यूब की बबिी लागत का अिुमाि लगाया गया था। 
सकिय िातु सामग्री को कम करके उत्प्रेरक की लागत को कम करिे का प्रयास ककया गया। 
20%Ni-1.7%Cu-1.7%Zn/Al2O3 उत्प्रेरक प्रदाि ककया गया >80% प्रारंसभक मीथेि रूपांतरण, 

र्ो 2h (>70%) तक CNTs (यािी, 18 gCNT gmetal-
1) की उच्च मािा के साथ श्स्थर रहा। 
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हाइड्रोर्ि उपर् और काबाि िैिोट्यूब विकास तंि पर िातु-समथाि बातचीत के प्रभाि का भी 
अध्ययि ककया गया था। 

 उत्प्रेरक के पुिर्ािि से बचिे के सलए उत्पाद्रदत काबाि िैिोट्यूब का उपयोग मीथेि 
अपघटि प्रनतकिया के सलए एक ऑटोकैटसलस्ट के रूप में भी ककया गया था। उसके सलए, 

60%Ni-5%Cu-5%Zn/Al2O3 और 30%Fe-2.5%Cu-2.5%Zn/Al2O3 उत्प्रेरक का परीक्षण 
750-950 डडग्री सेश्ल्सयस तापमाि रेंर् में ककया गया था। 900 डडग्री सेश्ल्सयस पर, 50 ग्राम 
सीएिटी के र्माि के बाद, 60%Ni-5%Cu-5%Zn/Al2O3 उत्प्रेरक िे बबिा ककसी निश्ष्ियता 
के 40 घंटे के सलए 30% निरंतर रूपांतरण प्रदाि ककया। 30%H2-70%CH4 (हाइथेि गैस कहा 
र्ाता है) के साथ उत्पाद गैस का उपयोग ऑटोमोद्रटि अिुप्रयोग के सलए संभावित ईंिि के 
रूप में ककया र्ा सकता है। एक ऑटोकैटसलस्ट के रूप में उत्पाद्रदत काबाि ििैोट्यूब का उपयोग 
करते समय, काबाि के एक िए रूप (यािी, ग्राफीि ग्राफ्टेड काबाि िैिोट्यूब) की पहचाि की 
गई थी। FESEM, EDX मैवपगं, HRTEM, रमि स्पेक्रोस्कोपी और TGA र्ैसी विसभन्ि 
विश्लेषणात्मक तकिीकों द्िारा इसके विकास तिं और भौनतक गुणों का पता लगाया गया। 
यह देखा गया कक ग्रेफीि ग्राफ्टेड सीएिटी उत्पाद्रदत काबाि िैिोट्यूब की तलुिा में अत्यधिक 
किस्टलीय और अधिक तापीय श्स्थर होते हैं। इसके अलािा, मौरू्दा सीएिटी पर मीथेि अपघटि 
तंि का अध्ययि ककया गया और विसभन्ि विश्लेषणात्मक उपकरणों द्िारा समझाया गया। 
उत्पाद्रदत काबाि िैिोट्यूब का उपयोग अल्रा-लाइट, लचीली, अत्यधिक झरझरा एयरर्ेल सामग्री 
के उत्पादि के सलए ककया गया था। तयैार सससलका-सीएिटी-सीएमसी एयरर्ेल एक तेल सोखिा 
अिुप्रयोग में उपयोग ककया गया था, और 1 ग्राम काबाि िैिोट्यूब एयररे्ल 28 ग्राम तेल को 
सोख सकता है। 
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